Title: Hypoxia-inducible factor-1 alpha, in association with inflammation, angiogenesis and MYC, is a critical prognostic factor in patients with HCC after surgery

Authors:

Chen-Xin Dai (daichenxin@gmail.com)
Qiang Gao (qgao1981@gmail.com)
Shuang-Jian Qiu (qiu.shuangjian@zs-hospital.sh.cn)
Min-Jie Ju (juminjie@hotmail.com)
Ming-Yan Cai (catcai@gmail.com)
Yong-Feng Xu (xuyongfengsh@gmail.com)
Jian Zhou (jianzhou99@yahoo.com)
Bo-Heng Zhang (bzhang@zshospital.com)
Jia Fan (jiafan99@yahoo.com)

Version: 3 Date: 23 November 2009

Author's response to reviews: see over
Dear Editor:

Thanks a lot for accepting our manuscript. Now, we have made all the changes as you told.

1. Major revisions (we require the author to make these changes)
   Additional files: Please do not use ranges when listing additional files. Please list all the additional files, for example: Additional files 4, 5, 6, 7, 8 and 9.
   **Revision:** We have changed from “additional files 4 to 9” to “Additional files 4, 5, 6, 7, 8 and 9” in last line of “Detailed legends” of “Table 5”.

2. Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above)
   I. Text: Please ensure that all text is in black
      **Revision:** We have changed some red words which marked changes in the first revision into black. And we make sure that all text in black now.
   II. Abstract: Please remove the keywords from the abstract.
      **Revision:** We have removed the keywords from the abstract.
   III. Additional files: if any figures or tables in the additional files are mentioned within the main text and additional file section please refer to them as Table S1, S2 or Figure S1, S2. This will avoid confusion with any figures and tables that are included within the main text.
      **Revision:** We have changed from “additional table 1, 2” or “additional figure 1, 2” to “Table S1, S2” or “Figure S1, S2” in both the main text and additional file section.
   IV. Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.
      **Revision:** We have checked the typography of our files carefully and correctly. It is very kind of you that you may change everything if it is needed so as to meet the formatting for publication. We really appreciated that we can share our
findings with other scientists through BMC Cancer which is a very important journal in the fields of biomedicine and cancer. Thank you very much!

With great regards,

Yours Sincerely,

Jia Fan, M.D., Ph.D., FACS
Prof. and Chief of Dept. of Liver Surgery
Vice President of Zhong Shan Hospital of Fudan University
136 Yi Xue Yuan Road
200032, Shanghai
P.R.China
Tel.& Fax.: +86-21-64037181